Author's response to reviews

Title: Perspectives on gastroesophageal reflux disease in primary care: the REFLEX study of patient-physician agreement

Authors:

Etienne Dorval (dorval@med.univ-tours.fr)
Jean-Francois Rey (jean-francois.rey@wanadoo.fr)
Christine Soufflet (Christine.Soufflet@astrazeneca.com)
Katarina Halling (khalling@patientreported.com)
Philippe Barthélemy (Philippe.Barthelemy@astrazeneca.com)

Version: 8 Date: 16 March 2011

Author's response to reviews:

Minor amends have been made as follows:
Page 5, paragraph 2: improved grammar
Page 7, paragraph 1: amended AXONAL to Axonal
Page 19, table 1: dagger symbol removed following 'Alcohol consumers'